Author's response to reviews

Title: Prevention of non-communicable disease in a population in nutrition transition: Tehran Lipid and Glucose Study Phase II (Study Protocols & Design)

Authors:

Fereidoun Azizi (Azizi@endocrine.ac.ir)
Arash Ghanbarian (ghanbarian@endocrine.ac.ir)
Amir Abbas Momenan (momenan@endocrine.ac.ir)
Farzad Hadaegh (fzhadaegh@endocrine.ac.ir)
Parvin Mirmiran (mirmiran@endocrine.ac.ir)
Mehdi Hedayati (hedayati@endocrine.ac.ir)
Yadollah Mehrabi (ymehrabi@gmail.com)
Saleh Zahedi-Asl (zahedi@endocrine.ac.ir)

Version: 4 Date: 30 September 2008

Author's response to reviews: see over
The Trials Editorial Team,

Dear Persons,

Attached please find revised manuscript MS: 3433352701616118 entitled "Prevention of non-communicable disease in a population in nutrition transition: Tehran Lipid and Glucose Study Phase II (Study Protocols & Design)". We have provided a list of abbreviations (page 3), a section entitled "Authors contributions" (page 2), a declaration of competing interest (page 1), a statement of approval of appropriate ethics committee (page 17), and compiled with the comments of the respected reviewer:

1. The aim of the study was added and the sample size calculation was removed from the abstract.

2. The analysis plans on page 4 was moved to statistical section.

3. The validation and references of FFQ method are stated in page 11.

4. Small editorial and typographical issues were considered.

We are in the progress of registering this trial in one of the two registers, as you have suggested before the publication of the manuscript and will inform you as soon as we obtain the registration number.

Best Regards,

Fereidoun Azizi, M.D.
Professor of Internal Medicine and Endocrinology
Director of Research Institute for Endocrine Sciences,
Shahid Beheshti University (M. C)
P.O. Box: 19395-4763
Tehran 19395, Iran
Phone: +98 21 2243 2500
Direct: +98 21 2240 9309
Fax: +98 21 2240 2463 or +98 21 2241 6264
Email: azizi@endocrine.ac.ir